Author's response to reviews

Title: Pharmacologic prophylaxis for atrial fibrillation following cardiac surgery: a systematic review

Authors:

Ioanna Koniari (iokoniari@yahoo.gr)
Efstratios Apostolakis (stratisapostolakis@yahoo.gr)
Christina Rogkakou (christinarog@hotmail.com)
Nikolaos G Baikoussis (ngbaik@yahoo.com)
Dimitrios Dougenis (ddougenis@med.upatras.gr)

Version: 3 Date: 11 October 2010

Author's response to reviews:

Cover letter – Journal of Cardiothoracic Surgery

Patras, October, 2010
To: the Editor-in-Chief
Journal of Cardiothoracic Surgery

Dear Sir or Madam,

We herewith submit the revised article entitled: "Pharmacologic prophylaxis for atrial fibrillation following cardiac surgery: a systematic review.

The manuscript was corrected in accordance with the Journal of Cardiothoracic Surgery reviewers. The majority of the corrections proposed-requested by the reviewers were done.

We thank very match Dr. Dedeilias, Dr Siminelakis and Dr Chronidou for their comments.

For Dr Mikroulis: 1. we mean 31 and 39% like he note.
2. we mean "two".

For Dr Kotoulas: commented on "the side effects of all these medications". We think that the side effect is important but is more chemistry and pharmacology rather than cardiology.

All our conclusions are referred at the end and not for any part or for any drug.
An English grammar revision was done. It is quite impossible to create tablets for all these information. Everything is written in the text. The style of our manuscript was changed in order to conform with the journal's style.

Sincerely